

Dear Editor,

Please find the enclosed edited manuscript in Word format (file name: NAFLD in 2014).

Title:

Nonalcoholic Fatty Liver Disease in 2014.

Author:

Monjur Ahmed.

Name of Journal: World Journal of Gastroenterology.

ESPS Manuscript No. 13454.

The manuscript has been improved according to the suggestions of reviewers:

1. More references have been added in brackets and superscripts with available doi.
2. Language has been polished.
3. Format has been updated.
4. Abstract and core tip have been added.

Thank you for publishing my manuscript in the *World Journal of Gastroenterology*.

Yours truly,

Monjur Ahmed

E-mail: monjur.ahmed@jefferson.edu

Answering to chief editor,

We had add tables as suggested.